**Author’s response to reviews**

**Title:** A Serious Adverse Drug Reaction Probably Induced by Clonazepam: a case report of Myotoxicity

**Authors:**

Xiaonian Han (xjtuhxn@126.com)

Jinping Wang (184232699@qq.com)

**Version:** 3  **Date:** 21 Oct 2019

**Author’s response to reviews:**

Dear Editors:

Thank you very much for your letter and for the comments concerning our manuscript entitled "A Serious Adverse Drug Reaction—Myotoxicity Probably Induced by Clonazepam" (ID: PHAT-D-19-00193). Those comments are all valuable and very helpful for revising and improving our paper. We have studied the comments carefully and have made correction which we hope meet with approval.

The point by point responses to the comments are as following:

1. Please remove the height and weight information of the patient, as this may compromise anonymity. If this information is needed with regards to the manuscript, please instead provide the information as ranges, for example 60-70kg and 160-170cm.

   Response: Thanks for the constructive suggestion. We have removed the height and weight information of the patient(Case presentation section, line1, page 2).

2. Please state in the title that this is a case report.

   Response: Thanks. We have stated that this is a case report in the title and changed the title of the article to “A Serious Adverse Drug Reaction Probably Induced by Clonazepam: a case report of Myotoxicity “. (The title section, page 1).

3. Please remove your 'Conclusions' heading, as this should be combined within the 'Discussions and Conclusions' section.
Response: Thanks. We have removed the 'Conclusions' heading (The Discussions and Conclusions section, page 6).

4. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Response: Thanks. We have represented authors' names using their full initials, not their full name in the Authors’ Contributions section (The Authors’ Contributions section, page 6).

5. Thank you for providing a response to the reviewers. As this document is no longer required at this stage of the publication process, please remove your ‘Authors response letter’ from your submission’s supplementary files.

Response: Thanks. We have removed our ‘Authors response letter’ from our submission’s supplementary files.

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Response: Thanks. We will upload our final manuscript as a single clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.

If you have any question about this paper, please don’t hesitate to let me know. Once again, thank you very much for your comments and suggestion.

Best regards,
Jinping Wang